

Pricing executive stock options under employment shocks

Julio Carmona, Ángel León and Antoni Vaello - Sebastiá

Ivie rking papers ס

Los documentos de trabajo del Ivie ofrecen un avance de los resultados de las investigaciones económicas en curso, con objeto de generar un proceso de discusión previo a su remisión a las revistas científicas. Al publicar este documento de trabajo, el Ivie no asume responsabilidad sobre su contenido.

Ivie working papers offer in advance the results of economic research under way in order to encourage a discussion process before sending them to scientific journals for their final publication. Ivie's decision to publish this working paper does not imply any responsibility for its content.

La Serie AD es continuadora de la labor iniciada por el Departamento de Fundamentos de Análisis Económico de la Universidad de Alicante en su colección "A DISCUSIÓN" y difunde trabajos de marcado contenido teórico. Esta serie es coordinada por Carmen Herrero.

The AD series, coordinated by Carmen Herrero, is a continuation of the work initiated by the Department of Economic Analysis of the Universidad de Alicante in its collection "A DISCUSIÓN", providing and distributing papers marked by their theoretical content.

Todos los documentos de trabajo están disponibles de forma gratuita en la web del Ivie http://www.ivie.es, así como las instrucciones para los autores que desean publicar en nuestras series.

Working papers can be downloaded free of charge from the Ivie website http://www.ivie.es, as well as the instructions for authors who are interested in publishing in our series.

Edita / Published by: Instituto Valenciano de Investigaciones Económicas, S.A.

Depósito Legal / Legal Deposit no.: V-3564-2009

Impreso en España (septiembre 2009) / Printed in Spain (September 2009)

WP-AD 2009-22

Pricing executive stock options under employment shocks^{*}

Julio Carmona, Ángel León and Antoni Vaello-Sebastiá**

Abstract

We obtain explicit expressions for the subjective, objective and market value of perpetual executive stock options (ESOs) under exogenous employment shocks driven by an independent Poisson process. Within this setup, we obtain the executive's optimal exercise policy which allows us to analyze the determinants of both, the subjective valuation by executives and the objective valuation by firms. The perpetual ESO is compared with the more realistic finite maturity ESO finding that the approximation is reasonably good. We also use the objective valuation's results for accounting purposes. Further, we analyze the objective valuation distribution when there is uncertainty about the employment shock parameter. Finally, the role of ESOs in the design of executive's incentives is also discussed.

Keywords: ESO, Risk Aversion, Undiversification, Incentives, FAS 123R.

JEL Classification: G11, G13, G35, M52

^{*} The authors are grateful for the financial support granted by the Instituto Valenciano de Investigaciones Económicas (Ivie). Angel León, Julio Carmona and Antoni Vaello-Sebastiá also acknowledge the financial support from the Spanish Ministry for Science and Innovation through the grants ECO2008-02599/ECO, SEJ 2007-64649/ECON and SEJ2007- 67895-C04-03 respectively. We are also grateful to an anonymous referee for helpful comments and suggestions. The contents of this paper are the sole responsibility of the authors.

^{**} J. Carmona and A. León: University of Alicante. A. Vaello-Sebastiá: University of Illes Balears. Corresponding author (Ángel León): aleon@ua.es.

1 Introduction

The increasing relevance of executive stock options (ESOs) as a component of corporate compensation has led the International Accounting Standard Board (IASB) to issue the International Financial Reporting Standard 2 (IFRS 2) in February 2004. In March 2004, the Financial Accounting Standard Board (FASB) has also revised the Financial Accounting Standard 123 (FAS 123R) with a similar purpose, namely, to provide a fair value method for shared based compensation arrangement.¹ Computing the fair value of one ESO is not straightforward, this is so because ESOs are American call options modified to create the required incentives for aligning executive's goals with shareholders' interests. Thus, ESOs cannot be sold or transferred, although partial hedge is possible by trading correlated assets. In addition, they can only be exercised after ending the vesting period. Consequently, standard methods for valuing American options are not directly applicable and a growing literature has been searching for a solution to the issue of ESO valuation. As in Hull and White (2004), Leung and Sircar (2009) and FASB statement 123R, we also assume that the possible dilution effect is anticipated by the the market and already reflected in the stock price immediately after the ESO grant.

As any American call option, the value of an ESO depends on its payoff and this is clearly determined by the holder's exercise policy. In this regard, the literature has distinguished among three possible values for ESOs. First, there is a subjective ESO value given by what one ESO is worth to an executive. This value is quite different from the market price obtained from standard risk neutral valuation methods, as the Black-Scholes price for European options. This difference is clearly illustrated by a common finding in the empirical literature on the subject. As illustrated by the works of Huddart and Lang (1996) and Bettis et al. (2005), ESOs are typically exercised well before maturity and mostly right after the end of the vesting period. In general, this behavior is inconsistent with the exercise policy that a well diversified investor facing an unrestricted environment would follow, giving rise to the ESO market price. Finally, motivated by the recent accounting standards, the literature has also introduced the objective value or the shareholders' cost when issuing the option. Although companies do not face the same hedging restrictions than their executives do, the ESO cost is not equal to its

 $^{^1\}mathrm{We}$ will only concentrate on the FAS 123R since both standards establish rather the same purpose concerning the fair value.

market price because of the executive's exercise policy.

Thus, either the subjective or the objective value turns out to be dependent on the precise way in which the exercise policy is incorporated. In this regard, we can distinguish two broad lines of analysis. In the first line, the exercise policy is obtained from structural models in which executives maximize their expected utility subject to a given set of constraints. Lambert et al. (1991) is an early example. They establish the subjective ESO value as its certainty-equivalence. Subsequently, Huddart (1994) and Kulatilaka and Marcus (1994) use this framework to provide an estimate of this subjective value determining the executive's exercise rule on a binomial tree. However, they restrict the executive to hold her wealth only in the risk free asset. Hall and Murphy (2002) and Cai and Vijh (2005) allow a more general setting in which the executive's wealth can also be held in restricted stocks or some market portfolio, respectively. Tian (2004) also uses this approach to study the incentive effects of European-style ESOs. All of them are static analyses in which the executive maximizes the expected utility of terminal wealth. Kahl et al. (2003) and Ingersoll (2006) use a dynamic approach such that the executive takes decisions about his optimal consumption and portfolio composition.

In a second line of reduced form models, the exercise policy is described either by some exogenous random event or by some exogenous parameter (or both) that forces the early option exercise and hence, the option expected life. An early example might be Jennergren and Näslund (1993), who introduce an exogenous and independent Poisson process whose first arrival forces the early exercise of (vested) options. Carpenter (1998) shows that this type of models performs as well as structural ones and Carr and Linetsky (2000) advocate their use because of their simplicity. Hull and White (2004) illustrate the inclusion of a virtual strike price that triggers early exercise and that can be calibrated so as to match the option expected life.

Our contribution can be placed along the first class of models. We use a simplified version of Ingersoll's (2006) model in which the executive allocates her wealth across the market portfolio and risk free bonds, but the risk factors are reduced to just the market risk. Therefore, the executive is constrained to hold more of the company stock than its corresponding share in the market portfolio.² As a result, there are two sources of risk, one coming from the non diversifiable systematic risk factor, underlying the market portfolio, and the other coming from the idiosyncratic component not correlated with market risk. In a well diversified portfolio, the only source of risk would come exclusively from the

 $^{^{2}}$ See the introduction to Ingersoll's (2006) article for a list of possible reasons of such undiversified holdings.

market portfolio and any other idiosyncratic component would have vanished. From the maximization of a lifetime utility function, an stochastic discount factor (SDF) is found which, by including those two sources of risk, can be used for pricing ESOs. We extend this model by including a job termination risk along the lines of Jennergren and Näslund (1993). The inclusion of this additional source of risk leads to a valuation model similar to that of Sircar and Xiong (2007). There is nonetheless a basic difference, they solve their model under the risk neutral measure which does not consider the executive's exercise policy. Instead, we use the real measure and obtain a closed form solution for the executive's exercise policy. Under this approach it is possible to provide a closed form expression for both, the subjective and the objective values. The case analyzed by Sircar and Xiong (2007) is only a solution for the ESO market value, which can be obtained in our approach by simply assuming a risk neutral agent (or a well diversified agent).

The merit of this approach lies on its ability to become a reasonable approximation for real situations based on finite maturity ESOs, which are computationally more demanding. We show the bias size for this approximation to be fairly small. We also discuss the impact on the cost of ESOs when the firm is uncertain about how likely the executive will finish the employment relationships. To this end, the firm is endowed with a prior distribution for the job termination risk. Finally, we study the executive's incentives by analyzing the subjective ESO delta and vega.

Section 2 presents the theoretical results for the subjective valuation of ESOs. Section 3 discusses how the subjective valuation is affected by changes in the relevant parameters and shows the size of the bias incurred by the use of the perpetual ESO against the finite maturity case. Section 4 includes the theoretical results about the objective ESO valuation or firm cost, the corresponding goodness of fit in the approximation to the finite maturity case is also discussed and the implications for the accounting standards. This section concludes with a discussion of the impact that uncertainty on the job termination risk has on the cost of ESO. In Section 5, we show how ESO affects executive's performance through his subjective 'greeks'. Section 6 concludes.

2 Subjective ESO valuation

Our benchmark model will be a perpetual ESO with a stochastic live arising from exogenous employment shocks which forces the termination of the employment relationship, as in the model of Jennergren and Näslund (1993). These shocks can arise from either the executive's side, due to voluntary resignation, or from the firm's side, due to the executive dismissal. In any case, the executive is forced to exercise the option if the event occurs after the vesting period, or to forfeit it, if the vesting period has not ended yet. The time at which the employment relationship is terminated is simply modeled as the first event of a Poisson process with hazard rate of λ per unit time. This Poisson process is assumed to be independent of any other stochastic process underlying our menu of assets. The hazard rate leads to jumps in the ESO price, but not in the underlying stock price, as in Jennergren and Näslund (1993). We assume that the job termination risk is not priced, so that it can be diversified away. This assumption is very common in the literature. See, for instance, Jennergren and Näslund (1993), Carpenter (1998), Carr and Linetsky (2000), Hull and White (2004), Sircar and Xiong (2007) and Leung and Sircar (2009).

Those employment shocks force exogenous exercise of ESOs. Endogenous exercise results from the optimizing behavior of the executive which typically is assumed to solve a constrained maximization problem. We shall consider the same menu of assets as in Ingersoll (2006), a risk-free bond, the market portfolio and the company stock. The equations describing the dynamics of the company stock and the market portfolio prices will be given by

$$\frac{dS}{S} = (\mu_S - q_S)dt + \sigma_S dZ_S ,$$

$$\frac{dM}{M} = (\mu_M - q_S)dt + \sigma_M dZ_M ,$$

where μ_s and μ_M denote the growth rate of the stock and the market portfolio respectively, while q_s and q_M stand for the corresponding continuous dividend yields. The firm's stock and the market portfolio are assumed to be imperfectly correlated. Formally, the Wiener processes satisfy the following relationship:

$$\sigma_S dZ_S = \beta \sigma_M dZ_M + \sigma_I dZ_I , \qquad (1)$$

where the parameter β is the conventional market beta, and dZ_M and dZ_I are independent standard Wiener processes. Notice that σ_I is not an independent parameter, since it must satisfy the restriction $\sigma_I^2 = \sigma_S^2 - \beta^2 \sigma_M^2$ as equation (1) makes clear. In short, we can write the equation for the stock price dynamics as

$$\frac{dS}{S} = (\mu_s - q_s) dt + \beta \sigma_M dZ_M + \sigma_I dZ_I .$$

As Ingersoll (2006), we assume that the executive is infinitely lived and maximizes an expected

lifetime utility of the constant relative risk aversion class. To capture the degree of undiversification, we define the parameter θ as the excess of company's stock holding over the optimal level already incorporated in the market portfolio.³ Therefore, the executive's problem is

$$\max_{C,\omega} \mathbb{E}_0 \left\{ \int_0^\infty e^{-\rho t} \frac{C_t^{1-\gamma}}{1-\gamma} dt \right\}$$
(2)

where $\mathbb{E}_0[\cdot]$ denotes the conditional expectation and it is subject to the following dynamic budget constraint:

$$dW = \{ [r + \omega (\mu_M - r) + \theta (\mu_S - r)] W - C \} dt + \omega \sigma_M W dZ_M + \theta \sigma_S W dZ_S , \qquad (3)$$

with initial condition $W(0) = W_0$. For simplicity, no wage income is assumed. Assuming that CAPM holds, i.e. $\mu_s = r + \beta (\mu_M - r)$, and using the orthogonal decomposition described in equation (1), we can rewrite equation (3) as

$$dW = \left\{ \left(r + \left(\omega + \theta \beta \right) \left(\mu_M - r \right) \right) W - C \right\} dt + \left(\omega + \theta \beta \right) \sigma_M W dZ_M + \theta \sigma_I W dZ_I .$$
(4)

Given the above conditions, we can obtain the SDF in the following lemma.

Lemma 1 The SDF, Θ , that prices the derivative will obey the following ordinary differential equation (ODE):

$$\frac{d\Theta}{\Theta} = -\hat{r}dt - \left(\frac{\mu_M - r}{\sigma_M}\right)dZ_M - \gamma\theta\sigma_I dZ_I \tag{5}$$

where $\hat{r} = r - \gamma \theta^2 \sigma_{I}^2$.

This proof is straightforward by following Ingersoll (2006). Notice that, when the executive is either risk neutral, $\gamma = 0$, or has a well diversified portfolio, $\theta = 0$, the SDF does not include any term reflecting the (diversifiable) idiosyncratic risk of the stock. Hence, the resulting value coincides with the risk neutral price for marketable options.

³Let $\underline{\theta}$ denote the minimum amount of the company stock that the executive is constrained to hold. If ξ^* denotes the optimal share of the company stock in the market portfolio and ω^* the optimal share of the market portfolio in the executive's total portfolio, then θ would satisfy the following condition $\theta = \underline{\theta} - \omega^* \xi^* \ge 0$.

2.1 Pricing without vesting period

For pricing ESOs, we use the no arbitrage condition $\mathbb{E}_0[d(\Theta V)] = 0$ where V denotes the perpetual ESO value. Thus,

$$0 = \mathbb{E}_0 \left[d(\Theta V) \right] = \mathbb{E}_0 \left[d(\Theta V) \right] \text{ no employment shock} \left(1 - \lambda dt \right)$$

+ $\mathbb{E}_0 \left[d(\Theta V) \right] \text{ employment shock} \left[\lambda dt \right].$ (6)

Given equation (6), we get the following result:

Lemma 2 Under the no arbitrage condition in equation (6) and assuming that CAPM holds, we get the following fundamental ODE for the subjective perpetual ESO price:

$$\left(\frac{\sigma_s^2}{2}\right)V_{SS}S^2 + \left(\hat{r} - \hat{q}_s\right)V_sS - \left(\hat{r} + \lambda\right)V + \lambda\Psi(S) = 0 , \qquad (7)$$

where $\hat{r} = r - \gamma \theta^2 \sigma_I^2$, $\hat{q}_S = q_S + \gamma \theta (1 - \theta) \sigma_I^2$ and $\Psi(S)$ denotes the payoff of the ESO holder if there is an employment shock defined as $\Psi(S) = (S - K) \mathbb{1}_{\{S > K\}}$, where $\mathbb{1}_{\{A\}}$ is the indicator function such that $\mathbb{1}_{\{A\}} = 1$ if A is true and $\mathbb{1}_{\{A\}} = 0$, otherwise.

Proof.- See Appendix A.

Equation (7) is the ODE defining the executive's value of the ESO in the continuation or waiting region. The structure of the problem implies that there is a threshold price, S^* , such that the optimal policy is waiting while $S < S^*$ and exercising as soon as $S \ge S^*$. Hence, at the boundary with the exercise region, the following conditions must hold:

$$V(S^*) = S^* - K, (8)$$

$$V'(S^*) = 1, (9)$$

where the threshold price S^* is determined endogenously as part of the complete solution. Considering equations (7), (8) and (9), we obtain the following proposition:

Proposition 3 Assume that there is no vesting period and the ESO is a perpetual American call option, then the solution to the ODE defined in (7) subject to the following boundary conditions, V(0) = 0 and equations (8) and (9), provides the value for the ESO holder, whose explicit solution is given by

$$V(S) = \begin{cases} \hat{A}_{1}K^{1-\hat{\alpha}_{1}}S^{\hat{\alpha}_{1}} & \text{if } S \leq K \\ \hat{B}_{1}K^{1-\hat{\alpha}_{1}}S^{\hat{\alpha}_{1}} + \hat{B}_{2}K^{1-\hat{\alpha}_{2}}S^{\hat{\alpha}_{2}} + \lambda \left(\frac{S}{\lambda + \hat{q}_{s}} - \frac{K}{\lambda + \hat{r}}\right) & \text{if } K < S \leq \hat{S}^{*} \\ S - K & \text{if } S > \hat{S}^{*} \end{cases}$$
(10)

where $\hat{\alpha}_1$ and $\hat{\alpha}_2$ are the solutions to the quadratic equation $(\sigma_s^2/2)\hat{\alpha}^2 + (\hat{r} - \hat{q}_s)\hat{\alpha} - (\hat{r} + \lambda)$ and the values of the constants \hat{A}_1 , \hat{B}_1 and \hat{B}_2 are defined in Appendix B by equations (27), (26) and (22) respectively. Finally, the threshold price \hat{S}^* is uniquely defined by solving the next equation:

$$\lambda \left(\frac{\hat{S}^*}{K}\right)^{\hat{\alpha}_2} = -\left(1 - \hat{\alpha}_2\right)\hat{r} - \hat{\alpha}_2\hat{q}_s\left(\frac{\hat{S}^*}{K}\right).$$
(11)

Proof.- See Appendix B.

V(S) is homogeneous of degree one in both S and K. The first two rows of equation (10) show the subjective ESO value when the price is below the optimal subjective threshold. Both belong to a situation in which the executive is better-off waiting rather than exercising the option. Consider the following intuitive explanation of these gains for the second waiting region. Its first component comes from the possible increase in the future price of the underlying stock. The second one concerns the possibility of exercising the ESO if an employment shock occurs at any future time with a probability of λ . Hence, the term $\lambda \left(S/(\lambda + \hat{q}_S) - K/(\lambda + \hat{r})\right)$ denotes the expected ESO present value when it is in-the-money.⁴ Of course, in the first waiting region this term does not appear since the option is out-of-the-money.

Some remarks about equation (11) are in order. By contrast with the paper of Sircar and Xiong (2007), we have found a closed form expression for the executive's exercise policy which is also valid for non-risk neutral executives. The exercise policy is homogeneous of degree one in K, that is, any change in the strike price implies a change in the same proportion in the threshold price. Finally, although it is not easy to find general comparative static results, all numerical simulations have provided results

$$\int_0^\infty \lambda e^{-\lambda t} \left[e^{-rt} \left(S e^{(r-q_S)t} - K \right) \right] dt$$

⁴This expected present value is obtained by solving the following integral:

where the time for the first employment shock follows an exponential distribution with mean $1/\lambda$.

that conform with intuition. For instance, a higher value of the employment shock probability tends to reduce the threshold price.

Figure 1 shows the typical shape of V(S) derived in Proposition 3. This figure displays several values for the stock excess holdings. Note that the situation of $\theta = 0$ is equivalent to the ESO risk neutral valuation. This value acts as an upper boundary for other situations in which the executive is less diversified.

2.2 Pricing with vesting period

In this subsection, we extend the results of Proposition 3 to the case of a positive vesting period, ν . The subjective ESO value at the granting date, t = 0, is obtained in the following proposition.

Proposition 4 The subjective ESO value at the granting date, t = 0, with a vesting period of length ν is given by

$$V_0^{SUB} = e^{-\lambda\nu} \mathbb{E}_0 \left[\frac{\Theta_\nu}{\Theta_0} V(S_\nu) \right]$$
(12)

where the exponential term is the probability that the executive will remain employed till the end of the vesting period and

$$\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}V(S_{\nu})\right] = e^{-\hat{r}\nu}\mathbb{E}_{0}\left[\left(S_{\nu}-K\right)\mathbb{1}_{\{S_{\nu}>\hat{S}^{*}\}}\right] + e^{-\hat{r}\nu}\mathbb{E}_{0}\left[\left(\hat{B}_{1}K^{1-\hat{\alpha}_{1}}S_{\nu}^{\hat{\alpha}_{1}}+\hat{B}_{2}K^{1-\hat{\alpha}_{2}}S_{\nu}^{\hat{\alpha}_{2}}+\lambda\left(\frac{S_{\nu}}{\lambda+\hat{q}_{s}}-\frac{K}{\lambda+\hat{r}}\right)\right)\mathbb{1}_{\{K< S_{\nu}\leq\hat{S}^{*}\}}\right] + e^{-\hat{r}\nu}\mathbb{E}_{0}\left[\left(\hat{A}_{1}K^{1-\hat{\alpha}_{1}}S_{\nu}^{\hat{\alpha}_{1}}\right)\mathbb{1}_{\{S_{\nu}\leq K\}}\right]$$
(13)

for any real numbers a, b and c verifying

$$\mathbb{E}_0\left[S_{\nu}^c \mathbb{1}_{\{a \le S_{\nu} \le b\}}\right] = \exp\left\{c\mu + c^2 \frac{\sigma^2}{2}\right\} \times \left\{\Phi\left(\frac{\ln b - \mu - c\sigma^2}{\sigma}\right) - \Phi\left(\frac{\ln a - \mu - c\sigma^2}{\sigma}\right)\right\}$$

where $\mu = \ln S_0 - (\hat{r} - \hat{q}_s - \sigma_s^2/2)\nu$, $\sigma = \sigma_s \sqrt{\nu}$ and $\Phi(\cdot)$ represents the cumulative standard normal distribution.

Proof.- See Appendix C.

3 Discussion

In this section, we begin studying the impact on the ESO valuation due to changes in several parameters of interest. These results can be found in the subsection of sensitivity analysis. We turn next to examine the robustness of our perpetual ESO valuation by comparing with American-style ESOs with finite maturities.

For all situations, we assume the risk free interest rate r = 6%, the continuous dividend yield $q_s = 1.5\%$ and the market volatility $\sigma_M = 20\%$. All these parameter values are taken on a yearly basis. We assume ESOs are granted at the money. The strike price, K, and the stock price at the granting date, S_0 , are equal to \$30.

3.1 Sensitivity analysis

We illustrate the effect of varying each of the following parameters: (i) the market beta, β , which can be either 0 or 1; (ii) the total yearly volatility of the stock return, σ_s , which can be 30%, 40% or 60%; (iii) the vesting period, ν , which can be either 0 or 3 years; (iv) the employment shock captured through the yearly Poisson intensity parameter, λ , with values of either 10% or 20%; (v) the excess stock holding, θ , ranging from 0% to 40%; and finally, (vi) the risk aversion parameter, γ , with value of 2 or 4. The results of this analysis are displayed in Table 1. This table is divided in four panels labelled from A to D according to different values of λ and ν . Namely, Panel A ($\lambda = 10\%$, $\nu = 0$), Panel B ($\lambda = 10\%$, $\nu = 3$), Panel C ($\lambda = 20\%$, $\nu = 0$) and Panel D ($\lambda = 20\%$, $\nu = 3$). Next, we comment on several features of interest.

First, the ESO market price, V^{RN} , is obtained under the restriction of $\theta = 0\%$. It is shown that V^{RN} increases with the total volatility of the stock return. This feature is independent of the risk decomposition into common risk or beta and specific or idiosyncratic risk, σ_I . The reason is that a well diversified agent does not worry about the size of σ_I . Note that, as expected, V^{RN} acts as an upper boundary for V^{SUB} .

Second, the higher the value of β the higher V^{SUB} . This effect has already been addressed by Tian (2004). A value of $\beta = 0$ suggests that the market portfolio is useless for hedging the risk of large holdings of the company stock. Observe that under $\beta = 0$, it holds that $\sigma_s = \sigma_I$. Thus, for any level of both the excess stock holding and risk aversion, an increase in the size of beta such that σ_s is keeping constant (i.e., a displacement across any row of Table 1), implies a decrease in the specific risk which improves the executive's utility.

Third, the higher the risk aversion or the excess stock holding, the lower V^{SUB} . Fourth, the higher the employment shock intensity the lower V^{SUB} . Compare panels A and C when there is no vesting period, while panels B and D for the positive one. Fifth, the higher the vesting period the lower V^{SUB} . Compare panels A and B for $\lambda = 10\%$, or panels C and D for $\lambda = 20\%$.

Last but not least, increases in σ_s have ambiguous effects on V^{SUB} . This is due to the fact that, as β is kept constant, changes in σ_I^2 are one-to-one changes in σ_s^2 . The conventional analysis of American call options shows that a higher variance tends to increase V^{SUB} . If this increase is due to a higher idiosyncratic volatility, the adjusted interest rate falls and the adjusted dividend yield increases and, pursuing further the analogy with American calls options, these changes tend to decrease V^{SUB} . In Table 1 we see that V^{SUB} tends to fall as σ_I increases when either γ or θ are large. For instance, V^{SUB} increases with σ_I in the column of Panel A for the values of $\beta = 1$, $\gamma = 2$ and $\theta = 10\%$. By contrast, V^{SUB} decreases in the column of Panel A for the values of $\beta = 1$, $\gamma = 2$ and $\theta = 30\%$. There also some transitional cases in which V^{SUB} exhibits an U-shaped behavior or its inverse. For instance, the column of Panel A for the values of $\beta = 1$, $\gamma = 2$ and $\theta = 30\%$. There also some transitional cases in $\beta = 1$, $\gamma = 2$ and $\theta = 20\%$ illustrates the U-shaped case, and the column of Panel B for the values of $\beta = 0$, $\gamma = 2$ and $\theta = 20\%$, the inverted U-shaped case.

Notice, however, that a higher value of λ tends to reverse the former relationship, making V^{SUB} increasing in σ_I . However, this effect only appears for short vesting periods, compare the values in Panel A with those in Panel C.⁵ Therefore, for the typical length of vesting periods observed in practice, this effect does not seem to modify substantially the behavior of V^{SUB} with respect to changes in σ_I .

3.2 Perpetual versus finite maturities

It becomes interesting to analyze if ESOs having finite maturities are adequately approximated by perpetual ones. We calculate ESO prices with finite maturities using the least-squares Monte Carlo algorithm (LSMC henceforth) of Longstaff and Schwartz (2001). We simulate the risk neutral price process but replacing r and q_s respectively by \hat{r} and \hat{q}_s , as defined in Lemma 2.

⁵In additional results, not reported here, for a vesting period of one year, an increase in λ from 10% to 20% enlarges the set of cases for which V^{SUB} increases when σ_I increases.

The LSMC uses a backward induction procedure that works as follows. At maturity, the ESO is exercised if it is in-the-money, then the subjective value of the ESO is $V_T^{SUB} = (S_T - K)^+$.⁶ One period before, at $T - \delta t$ where δt is the length of one time step, on one hand there is a probability equal to $1 - e^{\lambda \delta t}$ to abandon the firm and the payoff of the ESO would be $(S_{T-\delta t} - K)^+$. On the other hand, with probability $e^{\lambda \delta t}$ the executive remains in the firm and thus, he must decide either to hold or to exercise voluntarily the ESO. The executive will exercise the ESO if $S_{T-\delta t} - K > e^{-\hat{r}\delta t} \mathbb{E}_{T-\delta t} \left[V_T^{SUB} \right]$, in this case the ESO value will be $S_{T-\delta t} - K$. Otherwise, the payoff will be the discounted expected one period ahead ESO value. Thus, the ESO value at any time t, such that $T > t > \nu$, is computed as

$$V_t^{SUB} = e^{-\lambda\delta t} \left[X_t \mathbb{1}_{\{S_t - K < X_t\}} + (S_t - K)^+ \mathbb{1}_{\{S_t - K \ge X_t\}} \right] + \left(1 - e^{-\lambda\delta t} \right) (S_t - K)^+$$

where $X_t = e^{-\hat{r}\delta t} \mathbb{E}_t \left[V_{t+1}^{SUB} \right]$ is the discounted expectation of the ESO value.⁷ The conditional expected ESO value is computed by least-squares such that for those paths in-the-money, the one period ahead ESO value is regressed over some basis functions of the current stock price. We work backwards until the vesting or grant date with this scheme. We use 20,000 paths simulated with monthly frequency⁸ and we take the average of 50 previous estimations using 25 different seeds plus the corresponding 25 antithetics.

Figure 2 exhibits alternative subjective ESO values without vesting period for different values of β and λ from Table 1. The values of the risk aversion and the total volatility are, respectively, $\gamma = 2$ and $\sigma_s = 30\%$. The remaining parameters are the same as in Table 1. The selected time to maturities, T, go from 5 years until 25 years, denoted as V_T^{SUB} . It is observed that ESO prices do not change from about 15 years until the end and considering alternative values of θ , see the graphics on the left. The graphics of relative biases, $(V^{SUB} - V_T^{SUB})/V_T^{SUB}$, where V^{SUB} are the perpetual ESO values from Table 1 displayed on the right lead to a decreasing pattern as T increases. Indeed, this positive bias tends to be lower than 10% for the benchmark American-style ESO with a maturity of ten years. In short, the benchmark ESO value is well approximated through the perpetual one.⁹ For the shortest maturity, T = 5, the largest bias is around 25% for $\lambda = 10\%$, while it decreases significantly to

⁶Note that $(S_T - K)^+$ is the same as $(S_T - K)\mathbb{1}_{\{S_T > K\}}$. ⁷Hull and White (2004) and Ammann and Seiz (2004) also introduce in the same way the exit rate for the backwards induction in their binomial tree models.

 $^{^{8}}$ Stentoft (2004) obtains that the LSMC method with 10 exercise points per year produce very accurate prices compared with the ones obtained using a binomial model with 50,000 time steps. He argues that more accurate prices are obtained when increasing the simulated paths or the number of basis functions used as regressors.

⁹It is available upon request, though not reported here, a table similar to Table 1 for the case of an ESO with T = 10years containing the ESO values underlying the four graphics on the left-hand side of Figure 2.

approximately 10% for $\lambda = 20\%$. Therefore, the larger the probability of employment shock, the better the approximation through the perpetual ESO.

4 ESO cost for firms

ESOs are extensively used as a payment scheme that may help firms in retaining and motivating its executives. As a result, an ever increasing part of the executive's compensation packages have taken the form of ESOs and this has motivated a discussion about the precise way in which this cost should be evaluated. There is not yet a consensus on this issue because of the lack of a general agreement about what should be the right model. Despite of this, there is a general agreement of the role played by the subjective value in the computation of the ESO cost for firms. This cost depends on the executive's exercise policy which is a function of his subjective ESO valuation. Since the firm is not restricted to hedge the risk associated with the stock options, the objective valuation will be a straightforward modification of the risk neutral one. That is, the ESO cost for the firm is the amount that would receive if the ESO were sold to a well diversified investor with the risk-averse executive's exercise policy, to which we should also add, facing exogenous employment shocks that may end the employment relationship.

4.1 Objective ESO valuation

We shall denote by V^{OBJ} the objective ESO value. It is immediate that the former definition implies that V^{OBJ} is the solution to the ODE described in equation (7) for the case of risk neutral agents (or, equivalently, well diversified agents). Recall that this case follows from setting the parameter γ , or either the parameter θ , to zero. In this case the adjusted risk free rate, \hat{r} , and the adjusted dividend yield, \hat{q}_s , become respectively r and q_s . For ease of comparison, the ODE corresponding to the risk neutral valuation is:

$$\left(\frac{\sigma_s^2}{2}\right)V_{SS}S^2 + (r - q_s)V_sS - (r + \lambda)V + \lambda\Psi(S) = 0, \tag{14}$$

where the boundary conditions are now given by

$$V(0) = 0$$
 (15)

$$V(\hat{S}^*) = \hat{S}^* - K, \tag{16}$$

such that \hat{S}^* is the executive's threshold price obtained in Proposition 3.¹⁰

We start getting the explicit expression for V^{OBJ} when there is no vesting. This turns out to be a slight modification of Proposition 3. The precise result is stated in the following corollary.

Corollary 5 Assuming that there is no vesting period and the ESO is a perpetual American call option, the objective ESO value is given by the solution to the ODE defined in equation (14) subject to the boundary conditions (15) and (16):

$$V(S) = \begin{cases} \tilde{A}_1 K^{1-\alpha_1} S^{\alpha_1} & \text{if } S \leq K \\ \tilde{B}_1 K^{1-\alpha_1} S^{\alpha_1} + \tilde{B}_2 K^{1-\alpha_2} S^{\alpha_2} + \lambda \left(\frac{S}{\lambda+q_s} - \frac{K}{\lambda+r}\right) & \text{if } K < S \leq \hat{S}^* \\ S - K & \text{if } S > \hat{S}^* \end{cases}$$
(17)

where \hat{S}^* is the executive's threshold price obtained in Proposition 3, α_1 and α_2 are the roots corresponding to the risk neutral case and the values of $\tilde{A_1}$, $\tilde{B_1}$ and $\tilde{B_2}$ are defined in Appendix D by equations (34), (33) and (32), respectively.

Proof.- See Appendix D.¹¹

This result can also be extended to the case of a positive vesting period. Specifically, by plugging both the risk neutral SDF and the executive's threshold price, \hat{S}^* , into Proposition 4 we get the next corollary.

¹⁰Notice that V^{OBJ} can also be obtained as the solution to a perpetual up-and-in barrier option with \hat{S}^* as the upper barrier. See Ingersoll (2006) for the finite maturity case. ¹¹As in the subjective valuation case, V(S) is linearly homogeneous in S and K.

Corollary 6 The objective ESO cost is given by

$$V_{0}^{OBJ} = e^{-\lambda\nu} \left\{ e^{-r\nu} \mathbb{E}_{0} \left[(S_{\nu} - K) \mathbb{1}_{\{S_{\nu} > \hat{S}^{*}\}} \right] + e^{-r\nu} \mathbb{E}_{0} \left[\left(\tilde{B}_{1} K^{1-\alpha_{1}} S^{\alpha_{1}} + \tilde{B}_{2} K^{1-\alpha_{2}} S^{\alpha_{2}} + \lambda \left(\frac{S_{\nu}}{\lambda + q_{s}} - \frac{K}{\lambda + r} \right) \right) \mathbb{1}_{\{K < S_{\nu} \le \hat{S}^{*}\}} \right] + e^{-r\nu} \mathbb{E}_{0} \left[\left(\tilde{A}_{1} K^{1-\alpha_{1}} S^{\alpha_{1}}_{\nu} \right) \mathbb{1}_{\{S_{\nu} \le K\}} \right] \right\}$$
(18)

where the constants \tilde{A}_1 , \tilde{B}_1 and \tilde{B}_2 are those obtained in Corollary 5.

Proof.- The above equation follows easily from equation (13) of Proposition 4 by setting $\gamma = 0$, or $\theta = 0$, and taking \hat{S}^* in Proposition 3 as the threshold price.

As in Section 3.1 we can show, with the help of the former results, the effects on V^{OBJ} of changes in several parameters of interest. Figure 3 summarizes the main findings concerning the impact on V^{OBJ} of different values for γ , θ and β . We take the values of $\sigma_s = 30\%$, $\sigma_M = 20\%$, $\nu = 3$ and $\lambda = 20\%$ as our benchmark values. The straight line at the top of the figure represents $V^{RN} = 6.240$ which corresponds to the second row of Panel D in Table 1. The picture of the left hand side shows that V^{OBJ} decreases with the degree of risk aversion, γ , and with the undiversification parameter, θ . Conversely, a higher value of β implies, for given total volatility of the company stock and the market portfolio, a lower idiosyncratic volatility and hence, a higher value of V^{OBJ} . The intuition is that all those changes reduce the threshold price that triggers the ESO exercise. The picture on the right hand side displays the ratio between the subjective and the objective value. We can see that, although the former changes affect V^{SUB} and V^{OBJ} in the same way, the ESO ratio decreases significantly with the degree of risk aversion, γ , the undiversification parameter, θ and the size of the idiosyncratic risk, σ_I (displayed in the figure as a lower value for β). This is in line with the results of Tian (2004), Ingersoll (2006) and Chung, Hu and Fu (2008) for the case of European ESOs.

This fall in the ratio is explained by the fact that, although all these changes affect both, the impact on V^{OBJ} is only through a change in the threshold whereas the impact on V^{SUB} also includes the changes in \hat{r} and \hat{q}_s , which are absent in the former because it is computed under the risk-neutral measure (see either Corollary 5 or 6).

Finally, we have also explored the effect of different values of the intensity rate. As expected, V^{OBJ} decreases with higher values of λ and this effect turns out to be larger, the larger the ESO

vesting period. The ratio V^{SUB}/V^{OBJ} moves up only slightly with higher values of λ and they are not reported here to save space (but see Table 2 for an illustration of the impact on V^{OBJ} of higher values of λ).

4.2 Perpetual versus finite maturity

In this section, we perform the same exercise made in subsection 3.2 and compare the objective perpetual ESO valuation, V^{OBJ} , with that corresponding to the finite maturity case, V_T^{OBJ} . We consider a maturity of ten years, which is frequently used in practice. Additionally, in subsection 3.2, the subjective perpetual ESO value turns out to perform quite well for this maturity. Thus, it seems natural to use this maturity to check how well our valuation method approximates the finite maturity case. This comparison is exhibited in Table 2. Both perpetual and finite maturity ESOs are displayed on the left and right hand-sides respectively for alternative parameter values of λ , β , γ and θ and different vesting periods (one, three and five years in panels A to C respectively). In this table, we have only considered the case of $\sigma_s = 40\%$. Other values also yield similar results.

Inspection of Table 2 shows that the difference between both ESO values is not large and decreases for larger values of γ , θ and λ . For instance, it is shown that relative biases, measured as $(V^{OBJ} - V_T^{OBJ})/V_T^{OBJ}$, go from about 10% to 0% when θ increases for the case of $\lambda = 10\%$ and $\gamma = 2$. This bias can be negative with respect to the former situation for large values of γ , though the size is nearly the same as above. The case of $\lambda = 20\%$ and $\gamma = 4$ also exhibits positive relative biases ranging from 5% to 0% as θ increases. Again, this bias becomes negative in some situations when θ is large. In short, we can conclude that the perpetual valuation is a good approximation due to the small bias size and its lower computational intensity.

4.3 Accounting implications

As a result of the increasing relevance of ESOs in the managers compensation packages and the need to converge with other international standards, the FASB has revised its statement N^o 123. The new FAS 123R requires firms to disclose the method used for estimating the grant-date fair value of their ESO compensation packages. Among the valuation techniques that the FAS 123R consider acceptable are both lattice and closed form models, such as the binomial model and the Black-Scholes-Merton formula, respectively. Since ESOs are typically exercised before maturity, the FAS 123R (paragraph A26) explicitly requires that this fair value be based on its expected term, or expected life, rather than its maturity term. Furthermore, this expected life must be disclosed by firms as part of the shareholders' available information (FAS 123R, paragraph A240). In general, this expected life must be estimated. Kulatilaka and Marcus (1994), among others, have warned against the use of historical data with the purpose of estimating the option expected term. This is so because the ESO expected life is linked to the stock market performance during the relevant period and this might lead to poor predictions. As an example of an acceptable method for estimating this parameter, the FAS 123R suggests using the estimated ESO fair value, as obtained from a lattice model, as an input of a closed form model, such as an augmented Black-Scholes-Merton model, from which the implied expected term can be calculated.

Now, we illustrate the usefulness of the perpetual option approximation to the real finite maturity option to estimate the ESO expected term. Specifically, we take the expected life of the objective finite maturity ESO which is implicit in the corresponding value of Table 2, and compare it with the implied expected life that one would obtain by using the objective perpetual ESO price as the value of the FAS 123 adjusted European call price. This value is given by,

$$V^{FAS} = exp(-\lambda\nu) \times BS(L)$$
⁽¹⁹⁾

where BS(L) denotes the Black-Scholes (1973) formula with a time to maturity equal to L and a vesting period of length ν .

Figure 4 displays the results of this comparison through four pictures for two different values of the risk of the employment shock, $\lambda = 10\%$ and $\lambda = 20\%$, and two values of the degree of risk aversion, $\gamma = 2$ and $\gamma = 4$. In all cases, the objective finite ESO has a term to maturity of ten years. The remaining parameters are r = 6%, $q_s = 1.5\%$, $\beta = 1$ and $\sigma_s = 40\%$ which are used for the evaluation of both, the objective perpetual and finite ESO. As it can be seen, the objective perpetual approximation is good whenever either the risk of the employment shock or the degree of relative risk aversion is high, or whenever the degree of portfolio diversification is low.

4.4 Uncertainty in employment shocks

We study the effects of uncertainty in the likelihood of an employment shock, λ , that is one of the main parameters in this work. We implement a simple exercise just to motivate this situation. Notice that the employment relationship may terminate by the side of either the firm or the executive because of some exogenous event, for instance, the executive finds out a better available job. Each part is more uncertain about the likelihood the other part attaches to this event. Assume that this uncertainty arises only from the executive's side. Further, we also assume the firm's manager ignores the precise value of λ , although he has some a priori probability distribution for it. Of course, this will also generate a probability distribution for the objective value.

For simplicity, we assume a Triangular distribution¹² for the values of λ . We consider three prior distributions for λ , that might be representative of the executive's outside opportunities. For instance, in a highly concentrated industry (few firms), the executive's employment alternatives would tend to be scarce, so that the probability of leaving his actual job would be lower than in less concentrated industries. These prior distributions are classified according to the skewness behavior capturing alternative industry concentration levels. An asymmetric distribution with positive (negative) skewness represents a higher probability mass for small (large) values of λ because the industry is highly (scarcely) concentrated. And finally, we also consider a symmetric distribution for an intermediate situation.

For each assumed distribution of λ a total of 10,000 values have been simulated. For the cases of positive, zero and negative skewness, the values of λ going from the first to the third quartiles are, respectively, [7.5%, 21%], [14.1%, 25.6%] and [18.6%, 32.2%].

Figure 5, containing four pictures denoted from I to IV, displays the box-plots for the objective values, V^{OBJ} , implied by the above three distributions under several combinations of values for the diversification restriction, θ , and the total volatility, σ_s . The pairs represented in figures I to IV are respectively: { $\sigma_s = 30\%, \theta = 10\%$ }, { $\sigma_s = 30\%, \theta = 40\%$ }, { $\sigma_s = 60\%, \theta = 10\%$ } and { $\sigma_s = 60\%, \theta = 40\%$ }. We select a value for β of one and a level of relative risk-aversion of four. Clearly, the implied distributions of V^{OBJ} coming from the distributions of λ with positive skewness (boxplots labeled as A) exhibit a higher median value than those distributions of λ with negative skewness (boxplots C). This is a clear effect of the negative correlation between the size of λ and V^{OBJ} . For instance, these

 $^{^{12}}$ See chapter 40 in Evans et al. (2000).

differences go from a median value of 8 (boxplot A) to 5 (boxplot C) in picture I. Note also that a higher volatility leads to a higher median for the objective value. This value is higher the lower is the restriction on the portfolio diversification.

5 Incentive effects

The literature has paid great attention to the optimal design of the executive compensation package as a way of affecting their incentives. In general, these compensation packages consist of a fixed cash component, a certain amount of restricted stock and ESOs. Typically, the role of ESOs on executives' incentives has been approached by examining the sign and size of the ESO greeks. See, for instance, Ingersoll (2006), Tian (2004) and Chang et al. (2008) among others. Of course, it is understood that the relevant ESO greeks are those coming from the subjective valuation, those related with executive's perception of incentives. In this regard, there are two ways in which ESOs may affect executives' incentives. First, as part of an agency problem, they align executives and shareholders interests in raising stock price. The perceived reward from acting this way is measured by the subjective delta greek, that is, the partial derivative of V^{SUB} with respect to the initial price S_0 . And second, when a new investment project is taken, the distribution of firm's total risk between the systematic component and the specific one will generally change. Given that they affect V^{SUB} very differently, there might be a moral hazard problem because of executives' risk taking behavior. However, the results of subsection 3.1 show that executives have strong incentives to reduce the firm's specific component and to increase the systematic component of total volatility. Although both results might seem interesting from the shareholders' point of view, they are not quite so. For the first implication, notice that shareholders do not face any restrictions to diversify their portfolios. The second one confirms the well known result in the literature that ESOs generate incentives to increase the systematic component of total risk since it raises the firm's expected return. See for instance, Tian (2004).¹³

Despite all this, there are at least two reasons to use option greeks in the discussion of how ESOs affect executives' incentives. In the first place, there is a gap between subjective and objective ESO valuation and, hence, there is a distinction between the incentives perceived by executives and the cost

 $^{^{13}}$ In all numerical simulations performed, we have found that the subjective vega for firm's specific risk is strongly negative when total risk is held constant, and that the subjective vega for systematic risk is positive. Although these results are obtained for perpetual American ESOs, they have not been reported in the main text, since they are analogous to those obtained for the finite European ESOs which are typically considered in the literature.

to the firm of providing those incentives. In particular, Chang et al. (2008) have considered the ratio between the subjective and the objective greek for finite European ESOs as a measure of the size of the perceived incentives per unit cost of the firm. Since our perpetual American ESOs approximates reasonably the finite case as shown in subsection 3.2, we extend their analysis to the case of perpetual American ESOs. Secondly, it turns out that the ratio between subjective and objective greeks might be of help in the design of the composition of the executive compensation package, provided the parameters are conveniently interpreted. We consider next each issue in turn.

Let us consider first the incentives to raise shareholders wealth as measured by the subjective delta, Δ^{SUB} . The firm's cost of providing these incentives are measured by the objective delta, Δ^{OBJ} . The ratio $\Delta^{SUB}/\Delta^{OBJ}$ is then the reward perceived by the executive per unit cost of the firm. A ratio below one suggests that the executive's incentives are lower than those intended by the firm. As Chang et al. (2008) have observed a ratio of 0.5, it means that the executive only perceives half of the incentive effects intended by shareholders. Hence, to provide the same level of incentives, he must be offered at least twice as many ESOs incurring in a higher cost.¹⁴ We show that $\Delta^{SUB}/\Delta^{OBJ}$ is typically below one when plotted as a function of S_0 . With regard to the impact of unemployment shocks, a higher value of λ reduces somewhat the size of this ratio.¹⁵

This is depicted in picture I of Figure 6. The ratio $\Delta^{SUB}/\Delta^{OBJ}$ has been computed numerically by evaluating both partial derivatives using $S_0 = K$ as a midpoint, e.g. for at-the-money ESOs. It turns out that this ratio is monotonically decreasing with θ and below one, except for the case $\theta = 0$ where subjective and objective valuations coincide. This result is not modified by the consideration of different values of λ . Hence, the maximum perceived incentives to increase firm's market value per unit cost are achieved when the executive does not hold any restricted stock, for at-the-money ESOs.

The former analysis can also be extended to examine executive's incentives for investments in risky projects. This is measured by the subjective systematic risk vega, denoted by Λ_{β}^{SUB} , that shows the change in V^{SUB} resulting from a unit change in market beta. The firm's cost of providing those incentives is measured analogously by Λ_{β}^{OBJ} . As before, the ratio $\Lambda_{\beta}^{SUB}/\Lambda_{\beta}^{OBJ}$, is represented in picture II of Figure 6 against θ for the case of at-the-money ESOs. The corresponding numerical partial derivatives have been evaluated using $\beta = 1$ as the midpoint. Now, the shape of the curve suggests the

 $^{^{14}}$ With decreasing marginal benefits, the amount of ESOs offered to executives should be more than twice the initial level. 15 These results are available upon request.

existence of a maximum value of the ratio for a certain value of θ denoted as θ^* . Furthermore, this maximum value turns out to be greater than one.

This issue has been further explored in Figure 7, in which we plot $\Lambda_{\beta}^{SUB}/\Lambda_{\beta}^{OBJ}$ as a function of θ for several values of the idiosyncratic volatility and of the employment shock likelihood. As it is shown, the higher the idiosyncratic component, σ_I , the lower the required amount of restricted stock, θ^* . A higher value of the likelihood of employment shocks, λ , leads to a rise in θ^* . By the other hand, the maximum value of the ratio is clearly decreasing.

6 Conclusions

In this paper we have examined how the valuation of perpetual ESOs can be achieved by using a stochastic discount factor derived from the constrained intertemporal optimization problem faced by the executive. Following Ingersoll (2006), this constraint comes from the obligation to hold a proportion of the company stock higher than the optimal one. We obtain a stochastic discount factor that, by pricing the risk free asset, the market portfolio and the firm's stock, can also price the option. In a first step, we obtain a closed form expression for the executive's exercise policy. From this, a closed formula for the subjective value of ESOs with and without a vesting period is obtained. Furthermore, shareholders' cost follows easily from the subjective threshold price that characterizes the executive's exercise policy.

Despite considering perpetual ESOs, our valuation method approximates reasonably well the more realistic finite maturity case. This is true for both, the subjective and the objective ESO price. Since the latter is the ESO cost for firms and this must be calculated as part of the firm financial statements, our approach can be useful for this purpose. Furthermore, in Section 4.3, we have checked that our approach can also approximate reasonably well the expected term of the ESO, which is also part of the information the firm must disclosure to their shareholders.

Finally, we have also examined the impact of ESOs on executives incentives, highlighting the role of the employment shock likelihood. In particular, we have found that a higher job turnover rate implies lower incentives for both raising the company market value and taking on projects with high correlation with market expected return. In this latter case, there appears to be a critical value of θ for

which the executive's perceived incentive per unit cost of the firm achieves a maximum. This maximum is decreasing with respect to the size of the idiosyncratic volatility.

Appendix A

Using Ito's lemma, the money value of the ESO, V, obeys the following stochastic differential equation:

$$dV = \left[\left(\mu_S - q_S \right) V_S S + \frac{\sigma_S^2}{2} V_{SS} S^2 \right] dt + \beta \sigma_M V_S S dZ_M + \sigma_I V_S S dZ_I$$

This equation together with equation (5) will lead to our fundamental equation (7). Indeed, by omitting terms of order higher than dt, we obtain:

$$\begin{split} \mathbb{E}_{0} [Vd\Theta] &= -(r - \gamma \theta^{2} \sigma_{I}^{2}) V \Theta dt \\ \mathbb{E}_{0} [\Theta dV] &= (\mu_{S} - q_{S}) V_{S} S \Theta dt + \frac{\sigma_{S}^{2}}{2} V_{SS} S^{2} \Theta dt \\ \mathbb{E}_{0} [d\Theta dV] &= -(\mu_{S} - r) V_{S} S \Theta dt - \gamma \theta \sigma_{I}^{2} V_{S} S \Theta dt \end{split}$$

where \mathbb{E}_0 denotes the conditional expectation operator under the real measure and the CAPM condition, $\mu_S = r + \beta(\mu_M - r)$, has been used to obtain the third equation. Now, by straightforward substitution, we get:

$$0 = \mathbb{E}_0 \left[V d\Theta + \Theta dV + d\Theta dV \right] \left(1 - \lambda dt \right) + \mathbb{E}_0 \left[V d\Theta + \Theta \left(\Psi(S) - V \right) + d\Theta \left(\Psi(S) - V \right) \right] \lambda dt.$$

Hence,

$$-(r-\gamma\theta^2\sigma_I^2)V + (r-q_S-\gamma\theta\sigma_I^2)V_SS + \frac{\sigma_S^2}{2}V_{SS}S^2 + \lambda\left[\Psi(S) - V\right] = 0.$$

Finally, by defining $\hat{r} = r - \gamma \theta^2 \sigma_I^2$ and $\hat{q}_S = q_S + \gamma \theta (1 - \theta) \sigma_I^2$ we obtain equation (7).

Appendix B

The solution to equation (7) can be easily shown to be

$$V(S) = \begin{cases} \hat{a_1} S^{\hat{\alpha_1}} & \text{if } S < K\\ \hat{b_1} S^{\hat{\alpha_1}} + \hat{b_2} S^{\hat{\alpha_2}} + \left(\frac{\lambda S}{\lambda + \hat{q}_S} - \frac{\lambda K}{\lambda + \hat{r}}\right) & \text{if } S \ge K \end{cases}$$
(20)

where $\hat{\alpha}_1$ and $\hat{\alpha}_2$ are, respectively, the positive and negative root of the quadratic equation $\hat{\alpha}^2 + (\hat{b} - 1)\hat{\alpha} - \hat{c}$, for

$$\hat{c} \equiv \frac{2}{\sigma_S^2} (\hat{r} + \lambda) = -\hat{\alpha}_1 \hat{\alpha}_2 \quad \text{and} \quad \hat{b} \equiv \frac{2}{\sigma_S^2} (\hat{r} - \hat{q}_S) = 1 - \hat{\alpha}_1 - \hat{\alpha}_2 .$$
(21)

In equation (7) the negative root has been eliminated in the region S < K by imposing the boundary condition V(0) = 0. The constants \hat{a}_1 and \hat{b}_2 can be solved in terms of \hat{b}_1 and K by using the usual conditions of value

matching and smooth pasting for S = K:

$$\begin{aligned} \hat{a_1} K^{\hat{\alpha_1}} &= \hat{b_1} K^{\hat{\alpha_1}} + \hat{b_2} K^{\hat{\alpha_2}} + \left(\frac{\lambda}{\lambda + \hat{q}_S}\right) \left(\frac{\hat{r} - \hat{q}_S}{\lambda + \hat{r}}\right) K\\ \hat{\alpha_1} \hat{a_1} K^{\hat{\alpha_1}} &= \hat{\alpha_1} \hat{b_1} K^{\hat{\alpha_1}} + \hat{\alpha_2} \hat{b_2} K^{\hat{\alpha_2}} + \left(\frac{\lambda}{\lambda + \hat{q}_S}\right) K \end{aligned}$$

The solution for \hat{b}_2 is:

$$\hat{b}_{2} = \frac{2\lambda}{\sigma_{S}^{2}} \left(\frac{1}{\hat{\alpha}_{2} (\hat{\alpha}_{2} - 1) (\hat{\alpha}_{1} - \hat{\alpha}_{2})} \right) K^{1 - \hat{\alpha}_{2}} \equiv \hat{B}_{2} K^{1 - \hat{\alpha}_{2}}$$
(22)

And for \hat{a}_1 :

$$\hat{a}_{1} = \hat{b}_{1} + \frac{2\lambda}{\sigma_{S}^{2}} \left(\frac{1}{\hat{\alpha}_{1} (\hat{\alpha}_{1} - 1)} \right) \frac{K^{1 - \hat{\alpha}_{1}}}{(\hat{\alpha}_{1} - \hat{\alpha}_{2})}$$
(23)

To determine the remaining constant, $\hat{b_1}$, and the threshold price, S^* , we use equations (8) and (9) to get:

$$\hat{b_1}\hat{S^*}^{\hat{\alpha_1}} + \hat{b_2}\hat{S^*}^{\hat{\alpha_2}} + \left(\frac{\lambda\hat{S^*}}{\lambda + \hat{q_S}} - \frac{\lambda K}{\lambda + \hat{r}}\right) = \hat{S^*} - K , \qquad (24)$$

$$\hat{\alpha_1}\hat{b_1}\hat{S^*}^{\hat{\alpha_1}} + \hat{\alpha_2}\hat{b_2}\hat{S^*}^{\hat{\alpha_2}} + \frac{\lambda\hat{S^*}}{\lambda + \hat{q}_S} = \hat{S^*} .$$
(25)

Solving first for $\hat{b_1}$ in equation (25) one gets:

$$\hat{b_1} = \frac{-\hat{\alpha_2}}{\hat{\alpha_1}} \hat{b_2} \left(\hat{S}^* \right)^{\hat{\alpha_2} - \hat{\alpha_1}} + \frac{1}{\hat{\alpha_1}} \left(1 - \frac{\lambda}{\lambda + \hat{q}_S} \right) \left(\hat{S}^* \right)^{1 - \hat{\alpha_1}} \cdot$$

Or, after substituting for \hat{b}_2 and simplifying

$$\hat{b_1} = \frac{1}{\hat{\alpha}_1} \left\{ \left(\frac{2\lambda}{\sigma_s^2}\right) \frac{1}{\left(1 - \hat{\alpha}_2\right) \left(\hat{\alpha}_1 - \hat{\alpha}_2\right)} \left(\frac{\hat{S}^*}{K}\right)^{\hat{\alpha}_2 - \hat{\alpha}_1} + \left(1 - \frac{\lambda}{\lambda + \hat{q}_s}\right) \left(\frac{\hat{S}^*}{K}\right)^{1 - \hat{\alpha}_1} \right\} K^{1 - \hat{\alpha}_1} \equiv \hat{B}_1 K^{1 - \hat{\alpha}_1} .$$
 (26)

Hence, we can write \hat{a}_1 as:

$$\hat{a}_{1} = \hat{B}_{1}K^{1-\hat{\alpha}_{1}} + \frac{2\lambda}{\sigma_{S}^{2}} \left(\frac{1}{\hat{\alpha}_{1}(\hat{\alpha}_{1}-1)}\right) \frac{K^{1-\hat{\alpha}_{1}}}{(\hat{\alpha}_{1}-\hat{\alpha}_{2})} \equiv \hat{A}_{1}K^{1-\hat{\alpha}_{1}}$$
(27)

so that, equation (10) in the main text is obtained.

Finally, by combining equations (24) and (25) we get the implicit equation for solving for \hat{S}^* which, by using the relations given in equation (21), can be written as equation (11) in the main text.

Appendix C

We want to solve the following conditional expectation:

$$\begin{split} \mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}V(S_{\nu})\right] &= \hat{A}_{1}K^{1-\hat{\alpha_{1}}}E_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}^{\hat{\alpha_{1}}}\mathbbm{1}_{\{S_{\nu}\leq K\}}\right] + \hat{B}_{1}K^{1-\hat{\alpha_{1}}}\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}^{\hat{\alpha_{1}}}\mathbbm{1}_{\{K\leq S_{\nu}\leq \hat{S}^{*}\}}\right] + \\ &+ \hat{B}_{2}K^{1-\hat{\alpha_{2}}}\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}^{\hat{\alpha_{2}}}\mathbbm{1}_{\{K\leq S_{\nu}\leq \hat{S}^{*}\}}\right] + \frac{\lambda}{\lambda+\hat{q}_{S}}\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}\mathbbm{1}_{\{K\leq S_{\nu}\leq \hat{S}^{*}\}}\right] - \frac{\lambda}{\lambda+\hat{r}}K\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}\mathbbm{1}_{\{K\leq S_{\nu}\leq \hat{S}^{*}\}}\right] + \\ &+ \mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}\mathbbm{1}_{\{S_{\nu}\geq \hat{S}^{*}\}}\right] - K\mathbb{E}_{0}\left[\frac{\Theta_{\nu}}{\Theta_{0}}\mathbbm{1}_{\{S_{\nu}\geq \hat{S}^{*}\}}\right] \,. \end{split}$$

Thus, all expectations take the general form $\mathbb{E}_0\left[\frac{\Theta_{\nu}}{\Theta_0}S_{\nu}^c\mathbb{1}_{\{a\leq S_{\nu}\leq b\}}\right]$ for c any given real number. Given the stochastic dynamics driving S_{ν}^c and (Θ_{ν}/Θ_0) we have explicit expressions for each one of them, namely:

$$S_{\nu}^{c} = S_{0}^{c} \cdot \exp\left\{c\left(\mu_{-}q_{S} - \frac{\sigma_{S}^{2}}{2}\right)\nu + c\sigma_{S}\sqrt{\nu}\varepsilon\right\}$$
$$\frac{\Theta_{\nu}}{\Theta_{0}} = \exp\left\{-\left(\hat{r} + \frac{1}{2}\left(\frac{\mu_{M} - r}{\sigma_{M}}\right)^{2} + \frac{\gamma^{2}\theta^{2}\sigma_{I}^{2}}{2}\right)\nu - \left(\frac{\mu_{M} - r}{\sigma_{M}}\right)\sqrt{\nu}\varepsilon_{M} - \gamma\theta\sigma_{I}\sqrt{\nu}\varepsilon_{I}\right\}$$

where ε , ε_M and ε_I are *independent standard normal variables* satisfying $\sigma_S \varepsilon = \beta \sigma_M \varepsilon_M + \sigma_I \varepsilon_I$. Then, the expectation we seek to solve is given by a double integral of the form:

$$\int_{\varepsilon_{M}} \int_{\varepsilon_{I}} S_{0}^{c} \exp\left\{c\left(\mu_{S}-q_{S}-\frac{\sigma_{S}^{2}}{2}\right)\nu+c\left(\beta\sigma_{M}\varepsilon_{M}+\sigma_{I}\varepsilon_{I}\right)\sqrt{\nu}\right\}\times\right.\\ \left.\exp\left\{-\left(\hat{r}+\frac{1}{2}\left(\frac{\mu_{M}-r}{\sigma_{M}}\right)^{2}+\frac{\gamma^{2}\theta^{2}\sigma_{I}^{2}}{2}\right)\nu-\left(\frac{\mu_{M}-r}{\sigma_{M}}\right)\sqrt{\nu}\varepsilon_{M}-\gamma\theta\sigma_{I}\sqrt{\nu}\varepsilon_{I}\right\}\times\right.$$
(28)
$$\phi(\varepsilon_{M})\phi(\varepsilon_{I})d\varepsilon_{M}d\varepsilon_{I}$$

where $\phi(\cdot)$ denotes the density function of a standard normal variable. Notice that the range of integration for ε_M and ε_I must be such that $a \leq S_{\nu} \leq b$.

Following Cochrane and Saa-Requejo (1999), we define the new variables:

$$\delta_1 = rac{eta \sigma_M arepsilon_M + \sigma_I arepsilon_I}{\sigma_S} \quad ; \quad \delta_2 = rac{\sigma_I arepsilon_M - eta \sigma_M arepsilon_I}{\sigma_S} \; .$$

Notice that δ_1 and δ_2 are too *independent standard normal variables*. By reversing the change we have the following expression in terms of ε_M and ε_I :

$$\varepsilon_M = rac{\beta \sigma_M \delta_1 + \sigma_I \delta_2}{\sigma_S} \quad ; \quad \varepsilon_I = rac{\sigma_I \delta_1 - \beta \sigma_M \delta_2}{\sigma_S} \; .$$

After substitution into equation (28), we get the following expression:

$$S_{0}^{c} \exp\left\{c\left(\mu_{S}-q_{S}-\frac{\sigma_{S}^{2}}{2}\right)\nu-\left(\hat{r}+\frac{1}{2}\left(\frac{\mu_{M}-r}{\sigma_{M}}\right)^{2}+\frac{\gamma^{2}\theta^{2}\sigma_{I}^{2}}{2}\right)\nu\right\}\times\right.$$
$$\int_{\delta_{1}}\int_{\delta_{2}} \exp\left\{\left[c\sigma_{S}-\left(\frac{\beta(\mu_{M}-r)+\gamma\theta\sigma_{I}^{2}}{\sigma_{S}}\right)\right]\sqrt{\nu}\delta_{1}\right\}\times\right.$$
$$\exp\left\{-\left[\left(\frac{(\mu_{M}-r)}{\sigma_{M}}\right)-\beta\sigma_{M}\gamma\theta\right]\frac{\sigma_{I}}{\sigma_{S}}\sqrt{\nu}\delta_{2}\right\}\phi(\delta_{1})\phi(\delta_{2})d\delta_{1}d\delta_{2}.$$
(29)

We turn next to the specification of the integration region for each of the new variables. Clearly, for $a \leq S_{\nu} \leq b$ we have the following boundaries in terms of the δ_1 variable:

$$\frac{\ln\left(a/S_0\right) - \left(\mu_S - q_S - \sigma_S^2/2\right)\nu}{\sigma_S\sqrt{\nu}} \le \delta_1 \le \frac{\ln\left(b/S_0\right) - \left(\mu_S - q_S - \sigma_S^2/2\right)\nu}{\sigma_S\sqrt{\nu}}$$

or more compactly $A \leq \delta_1 \leq B$. By the other hand the range of integration for δ_2 is unrestricted. Hence, by omitting the exponential term that appears outside the double integral (29), we are left with

$$\frac{1}{\sqrt{2\pi}} \int_{\delta_2} \exp\left\{-\left[\left(\frac{\mu_M - r}{\sigma_S}\right) - \beta \sigma_M \gamma \theta\right] \left(\frac{\sigma_I}{\sigma_S}\right) \sqrt{\nu} \delta_2 - \frac{1}{2} \delta_2^2\right\} d\delta_2 \times \frac{1}{\sqrt{2\pi}} \int_A^B \exp\left\{\left[c\sigma_S - \left(\frac{\beta(\mu_M - r) + \gamma \theta \sigma_I^2}{\sigma_S}\right)\right] \sqrt{\nu} \delta_1 - \frac{1}{2} \delta_1^2\right\} d\delta_1 \right\}$$

Each integral is solved by completing the square. Thus, for the first integral, we get

$$\exp\left\{\frac{1}{2}\left[\left(\frac{\mu_M-r}{\sigma_M}\right)-\beta\sigma_M\gamma\theta\right]^2\left(\frac{\sigma_I}{\sigma_S}\right)^2\nu\right\},\,$$

whereas for the second integral, we obtain:

$$\exp\left\{\frac{1}{2}\left[c\sigma_{S}-\left(\frac{\beta(\mu_{M}-r)+\gamma\theta\sigma_{I}^{2}}{\sigma_{S}}\right)\right]^{2}\nu\right\}\times\left\{\Phi\left[\frac{\ln\left(b\right)-\mu-c\sigma^{2}}{\sigma}\right]-\Phi\left[\frac{\ln\left(a\right)-\mu-c\sigma^{2}}{\sigma}\right]\right\},$$

for $\mu = \ln(S_0) + (\hat{r} - \hat{q}_s - (\sigma_s^2/2))\nu$ and $\sigma = \sigma_s \sqrt{\nu}$. In the computation of this integral we have made use of the relationship $\hat{r} - \hat{q}_s = r - q_s - \gamma \theta \sigma_I^2$ and the CAPM condition $(\mu_s - r) = \beta (\mu_M - r)$.

Finally and after some algebra, the product of the three remaining exponentials can be greatly simplified

$$\exp\left\{-\hat{r}\nu\right\}\exp\left\{c\mu+c^2\frac{\sigma^2}{2}\right\}$$

Summing up, we obtain:

 to

$$\mathbb{E}_{0}\left[\left(\frac{\Theta_{\nu}}{\Theta_{0}}S_{\nu}^{c}\right)\mathbb{1}_{\left\{a\leq S_{\nu}\leq b\right\}}\right] = \exp\left\{-\hat{r}\nu\right\}\exp\left\{c\mu + c^{2}\frac{\sigma^{2}}{2}\right\} \times \left\{\Phi\left(\frac{\ln b - \mu - c\sigma^{2}}{\sigma}\right) - \Phi\left(\frac{\ln a - \mu - c\sigma^{2}}{\sigma}\right)\right\}$$

Appendix D

Following the same steps as in Appendix C, we find that the solution to equation (14) is given by

$$V(S) = \begin{cases} \tilde{a_1} S^{\alpha_1} & \text{if } S < K\\ \tilde{b_1} S^{\alpha_1} + \tilde{b_2} S^{\alpha_2} + \left(\frac{\lambda S}{\lambda + q_S} - \frac{\lambda K}{\lambda + r}\right) & \text{if } K \le S \le \hat{S}^*\\ S - K & \text{if } S \ge \hat{S}^* \end{cases}$$
(30)

where α_1 and α_2 are, respectively, the positive and negative root of the quadratic equation $\alpha^2 + (b-1)\alpha - c$, for

$$c \equiv \frac{2}{\sigma_s^2}(r+\lambda) = -\alpha_1 \alpha_2 \quad \text{and} \quad b \equiv \frac{2}{\sigma_s^2}(r-q_s) = 1 - \alpha_1 - \alpha_2 .$$
(31)

Again, a similar procedure leads to the following values for the constants $\tilde{a_1}, \tilde{b_2}$ and \tilde{b}_1 :

$$\tilde{b}_{2} = \frac{2\lambda}{\sigma_{s}^{2}} \left(\frac{1}{\alpha_{2}(\alpha_{2}-1)(\alpha_{1}-\alpha_{2})} \right) K^{1-\alpha_{2}} := \tilde{B}_{2}K^{1-\alpha_{2}}$$

$$\tilde{b}_{1} = \left(1 - \frac{(2\lambda/\sigma_{s}^{2})}{(\alpha_{1}-1)(1-\alpha_{2})} \right) (\hat{S}^{*})^{1-\alpha_{1}} - \left(1 + \frac{(2\lambda/\sigma_{s}^{2})}{\alpha_{1}\alpha_{2}} \right) K(\hat{S}^{*})^{\alpha_{1}} - \tilde{b}_{2}(\hat{S}^{*})^{\alpha_{2}-\alpha_{1}} =$$

$$= \left\{ \left(1 - \frac{(2\lambda/\sigma_{s}^{2})}{(\alpha_{1}-1)(1-\alpha_{2})} \right) \left(\frac{\hat{S}^{*}}{K} \right)^{1-\alpha_{1}} - \left(1 + \frac{(2\lambda/\sigma_{s}^{2})}{\alpha_{1}\alpha_{2}} \right) \left(\frac{\hat{S}^{*}}{K} \right)^{\alpha_{1}} - \tilde{B}_{2} \left(\frac{\hat{S}^{*}}{K} \right)^{\alpha_{2}-\alpha_{1}} \right\} K^{1-\alpha_{1}}$$

$$\equiv \tilde{B}_{1}K^{1-\alpha_{1}}$$
(33)

$$\tilde{a}_1 = \tilde{b}_1 + \frac{2\lambda}{\sigma_S^2} \left(\frac{1}{\alpha_1(\alpha_1 - 1)} \right) \frac{K^{1-\alpha_1}}{(\alpha_1 - \alpha_2)} \equiv \tilde{A}_1 K^{1-\alpha_1} .$$
(34)

			$\beta = 0$						$\beta = 1$				
	θ	0.00	0.10	0.20	0.30	0.40		0.00	0.10	0.20	0.30	0.40	
γ	$\sigma_{\scriptscriptstyle S}$		Panel A: $\lambda = 0.1, \nu = 0$										
2	0.30	11.000	8.802	7.404	6.420	5.683		11.000	9.673	8.714	7.987	7.416	
	0.40	12.859	9.617	7.813	6.624	5.768		12.859	10.266	8.694	7.611	6.810	
	0.60	16.248	10.866	8.410	6.923	5.911		16.248	11.273	8.899	7.432	6.419	
4	0.30	11.000	7.273	5.415	4.281	3.516		11.000	8.605	7.127	6.105	5.348	
	0.40	12.859	7.644	5.456	4.215	3.416		12.859	8.527	6.453	5.190	4.332	
	0.60	16.248	8.174	5.538	4.185	3.345		16.248	8.656	6.007	4.596	3.719	
					Pε	anel B: λ	۱ =	= 0.1, ν =	3				
2	0.30	9.778	7.692	6.365	5.415	4.689		9.778	8.516	7.613	6.928	6.392	
	0.40	11.324	8.232	6.450	5.221	4.310		11.324	8.858	7.337	6.259	5.446	
	0.60	14.095	8.756	6.054	4.356	3.216		14.095	9.187	6.616	4.961	3.823	
4	0.30	9.778	6.233	4.382	3.169	2.311		9.778	7.505	6.098	5.101	4.343	
	0.40	11.324	6.259	3.892	2.466	1.560		11.324	7.159	5.012	3.617	2.645	
	0.60	14.095	5.733	2.636	1.205	0.533		14.095	6.296	3.193	1.635	0.828	
			Panel C: $\lambda = 0.2, \nu = 0$										
2	0.30	8.296	6.930	5.994	5.306	4.774		8.296	7.485	6.863	6.370	5.969	
	0.40	9.951	7.863	6.597	5.720	5.066		9.951	8.297	7.216	6.434	5.837	
	0.60	13.080	9.380	7.507	6.311	5.468		13.080	9.677	7.884	6.718	5.885	
4	0.30	8.296	5.920	4.612	3.766	3.170		8.296	6.803	5.807	5.086	4.536	
	0.40	9.951	6.492	4.856	3.865	3.197		9.951	7.116	5.610	4.639	3.952	
	0.60	13.080	7.340	5.179	3.990	3.238		13.080	7.716	5.573	4.360	3.574	
			Panel D: $\lambda = 0.2, \nu = 3$										
2	0.30	6.240	5.062	4.263	3.668	3.199		6.240	5.538	5.009	4.593	4.258	
	0.40	7.365	5.560	4.433	3.625	3.011		7.365	5.939	4.996	4.300	3.761	
	0.60	9.450	6.120	4.290	3.108	2.304		9.450	6.404	4.674	3.528	2.729	
4	0.30	6.240	4.192	3.019	2.215	1.630		6.240	4.954	4.100	3.470	2.978	
	0.40	7.365	4.320	2.743	1.758	1.120		7.365	4.894	3.495	2.550	1.878	
	0.60	9.450	4.077	1.901	0.874	0.389		9.450	4.464	2.296	1.184	0.602	

Table 1: Subjective perpetual ESO valuation

This table shows the subjective ESO value obtained by using either equation (10) or (13). The first column, γ , contains different risk-aversion coefficients. The second column, σ_S , contains the different levels of firm's stock volatility. The next five columns are obtained with $\beta = 0$ and the remaining ones with $\beta = 1$. We also consider different values of θ ranging from 0% to 40%. The table is divided in four panels for different combinations of the employment shock intensity, λ , and the vesting period, ν . Other parameters for ESO valuation are: $S_0 = K = \$30$, corresponding to the initial firm's stock price and the option strike price respectively, the yearly risk-free rate, r = 6%, the yearly stock continuously compounded dividend rate, $q_S = 1.5\%$ and the yearly market portfolio volatility, $\sigma_M = 20\%$.

				Perpetual ESO price					Finite maturity ESO price				
					θ					θ			
λ	β	γ	0.00	0.10	0.20	0.30	0.40	0.00	0.10	0.20	0.30	0.40	
							Panel	A: $\nu = 1$					
0.1	0	2	12.534	11.907	11.007	10.217	9.572	11.424	10.932	10.377	9.892	9.479	
		4	12.534	10.893	9.310	8.280	7.595	11.424	10.328	9.332	8.568	7.960	
	1	2	12.534	12.136	11.500	10.892	10.361	11.424	11.082	10.680	10.293	9.960	
		4	12.534	11.404	10.084	9.099	8.383	11.424	10.620	9.825	9.175	8.627	
0.2	0	2	9.339	9.002	8.481	8.011	7.622	8.429	8.609	8.249	7.938	7.659	
		4	9.339	8.420	7.479	6.840	6.398	8.429	8.207	7.573	7.070	6.672	
	1	2	9.339	9.130	8.766	8.407	8.087	8.429	8.702	8.446	8.190	7.977	
		4	9.339	8.716	7.940	7.345	6.900	8.429	8.407	7.899	7.475	7.116	
							Panel	B: $\nu = 3$					
0.1	0	2	11.324	10.795	10.175	9.701	9.342	10.217	9.919	9.664	9.452	9.253	
		4	11.324	10.103	9.201	8.665	8.318	10.217	9.645	9.197	8.794	8.460	
	1	2	11.324	10.975	10.500	10.102	9.784	10.217	9.981	9.780	9.623	9.478	
		4	11.324	10.435	9.625	9.089	8.718	10.217	9.778	9.432	9.110	8.821	
0.2	0	2	7.365	7.120	6.830	6.606	6.433	6.986	6.821	6.690	6.572	6.454	
		4	7.365	6.800	6.372	6.108	5.929	6.986	6.681	6.421	6.199	6.010	
	1	2	7.365	7.204	6.981	6.793	6.640	6.986	6.854	6.757	6.664	6.584	
		4	7.365	6.953	6.573	6.316	6.132	6.986	6.745	6.557	6.379	6.219	
							Panel	C: $\nu = 5$					
0.1	0	2	9.905	9.520	9.120	8.827	8.610	8.826	8.708	8.587	8.470	8.368	
		4	9.905	9.075	8.526	8.205	7.998	8.826	8.577	8.341	8.118	7.933	
	1	2	9.905	9.644	9.326	9.074	8.878	8.826	8.737	8.654	8.565	8.482	
		4	9.905	9.284	8.781	8.459	8.237	8.826	8.642	8.463	8.290	8.129	
0.2	0	2	5.465	5.319	5.168	5.056	4.972	5.125	5.070	5.011	4.954	4.901	
		4	5.465	5.152	4.942	4.814	4.728	5.125	5.006	4.889	4.785	4.698	
	1	2	5.465	5.366	5.245	5.149	5.073	5.125	5.081	5.042	4.999	4.959	
		4	5.465	5.231	5.040	4.915	4.826	5.125	5.036	4.949	4.868	4.790	

Table 2: Objective ESO valuation: perpetual vs. finite maturity (10 years)

This table shows the objective value for the perpetual ESO (left side) and for a finite maturity ESO of ten years (right side) denoted as V_T^{SUB} and V_T^{SUB} , respectively. The first column, λ , contains different values of the employment shock intensity. The second column, β , contains different levels of the market beta. The third column, γ , represents different levels of executive's risk aversion. V^{SUB} and V_T^{SUB} are obtained under five different values of θ ranging from 0% to 40%. We also consider three different vesting period lengths in Panel A ($\nu = 1$ years), Panel B ($\nu = 3$ years) and Panel C ($\nu = 5$ years). The perpetual values have been obtained using Corollary 6. For the finite maturity cases, the procedure to obtain these prices are described in subsection 3.2. The values for the remaining parameters are: $S_0 = K = \$30$, r = 6%, $q_S = 1.5\%$, $\sigma_S = 40\%$ and $\sigma_M = 20\%$.

Figure 1: Subjective ESO price and diversification level



The figure plots the subjective ESO price, V^{SUB} , as a function of the stock price for different values of the portfolio undiversification parameter, θ . Note that the continuous line for $\theta = 0$ represents the market ESO price, which acts as an upper bound for V^{SUB} . The values of the remaining parameters are: r = 6%, $q_S = 1.5\%$, $\gamma = 4$, $\lambda = 20\%$, $\sigma_M = 20\%$, $\sigma_S = 40\%$, K = \$30 and $\nu = 0$.



This figure shows the subjective ESO value (left-hand graphics) for different finite maturities (in x-axis) ranging from T = 5 to T = 25 years, denoted as V_T^{SUB} . Each graphic exhibits a different pair of values for (β, λ) and each line, inside each graphic, corresponds to a different value of θ . Any right-hand graphic displays the relative bias of the perpetual ESO computed as $(V_T^{SUB} - V_T^{SUB})/V_T^{SUB}$, where V^{SUB} is the perpetual ESO price from Table 1. The procedure to obtain V_T^{SUB} is described in Subsection 3.2. The values for the remaining parameters are: $S_0 = K = \$30, r = 6\%, q_S = 1.5\%, \lambda = 20\%, \sigma_S = 40\%, \sigma_M = 20\%$ and $\nu = 3$ years.



The left hand graphic represents the firm cost, V^{OBJ} , of a perpetual ESO for several values of γ and β . In this figure, the continuous line represents the risk neutral value, V^{RN} , of the perpetual ESO. The right hand graphic displays the ratio between the subjective and the objective values. The values of the other parameters are: r = 6%, $q_S = 1.5\%$, $\sigma_S = 30\%$, $\sigma_M = 20\%$, $\lambda = 20\%$, $S_0 = K = \$30$ and $\nu = 3$.





This figure compares the expected term from a finite maturity ESO of 10 years and a vesting period of 3 years (discontinuous line) with that obtained by calibrating equation (19) using a price equal to the perpetual ESO value (solid line). The remaining values of the parameters are: r = 6%, $q_S = 1.5\%$, $\sigma_S = 40\%$ and $S_0 = K = \$30$.



Figure 5: Box-plot for objective ESO prices

This figure shows the distribution for objective ESO values implied by the corresponding random samples drawn from alternative Triangular distributions of λ , denoted as A, B and C which have been described in subsection 4.4. The values for the remaining parameters are: r = 6%, $q_S = 1.5\%$, $\gamma = 4$, $\beta = 1$, $\sigma_S = 40\%$, $\beta = 1$, $S_0 = \$30$ and K = \$30.

Figure 6: Greek ratios and employment shock



Picture I displays the delta ratio, $\Delta^{SUB}/\Delta^{OBJ}$, as a function of θ taking $S_0 = K$ as a midpoint in the computation of the numerical partial derivatives. Picture II displays the systematic risk ratio, $\Lambda_{\beta}^{SUB}/\Lambda_{\beta}^{OBJ}$, as a function of θ using $\beta = 1$ as a midpoint. Each picture depicts several plots for different values of the likelihood of employment shocks, λ . The values for the remaining parameters are: r = 6%, $q_S = 1.5\%$, $\gamma = 2$, $\beta = 1$, $\sigma_M = 20\%$ and $\nu = 3$.





Pictures I and II display the systematic vega ratio, $\Lambda_{\beta}^{SUB}/\Lambda_{\beta}^{OBJ}$, as a function of θ for $\lambda = 5\%$ and $\lambda = 10\%$, respectively. In both cases, the ratio is computed for at-the-money ESOs and $\gamma = 2$. Each picture depicts several plots for different idiosyncratic risk values, σ_I . The values for the remaining parameters are: r = 6%, $q_S = 1.5\%$, $\beta = 1$, $\sigma_M = 20\%$ and $\nu = 3$.

References

- Ammann, M. and R. Seiz 2004. Valuing employee stock options: Does the model matter? Financial Analysts Journal 60(5):21–37.
- [2] Bettis, J. C., J. M. Bizjak, and M. L. Lemmon 2005. Exercise behavior, valuation, and the incentive effects of employee stock options. *Journal of Financial Economics* 76(2):445–470.
- Black, F. and M. Scholes 1973. The pricing of options and corporate liabilities. Journal of Political Economy 81(3):637–659.
- [4] Cai, J. and A. M. Vijh 2005. Executive stock and option valuation in a two state-variable framework. *Journal of Derivatives* 12(3):9–27.
- [5] Calvet, A. L. and A. H. Rahman 2006. The subjective valuation of indexed stock options and their incentive effects. *The Financial Review* 41:205–227.
- [6] Carpenter, J. N. 1998. The exercise and valuation of executive stock options. Journal of Financial Economics 48:127–158.
- [7] Carr, P. and V. Linetsky 2000. The valuation of executive stock options in an intensity-based framework. *European Finance Review* 4:211–230.
- [8] Chang, C., C. Der-Fuh, and Y.-H. Hsu 2008. ESO compensation: The roles of default risk, employee sentiment, and insider information. *Journal of Corporate Finance* 14:630–641.
- [9] Cochrane, J. H. 2005. Asset Pricing. : Princeton University Press.
- [10] Cochrane, J. H. and J. Saa-Requejo 2001. Beyond arbitrage: Good-deal asset price bounds in incomplete markets. *Journal of Political Economy* 108(1):79–119.
- [11] Cvitanic, J., Z. Wiener, and F. Zapatero 2008. Analytic pricing of employee stock options. *Review of Financial Studies* 21(2):683–724.
- [12] Dixit, A. and R. Pyndick 1994. Investment under Uncertainty. : Princeton University Press.
- [13] Evans, M., N. Hastings, and B. Peacock 2000. Statistical Distributions. : John Wiley and Sons.
- [14] Financial Accounting Standard Board 1995. Statement of financial accounting standards nº 123: Accounting for stock-based compensation. FASB.
- [15] Financial Accounting Standard Board 2004. Statement of financial accounting standards nº 123 (revised 2004): Share-based payments. FASB.
- [16] Hall, B. J. and K. J. Murphy 2000. Optimal exercise prices for executive stock options. American Economic Review 90:209–214.
- [17] Hall, B. J. and K. J. Murphy 2002. Stock options for undiversified executives. Journal of Accounting and Economics 33(1):3–42.

- [18] Henderson, V. 2005. The impact of the market portfolio on the valuation, incentives and optimality of executive stock options. *Quantitative Finance* 5(1):35–47.
- [19] Huddart, S. 1994. Employee stock options. Journal of Accounting and Economics 18(2):207–231.
- [20] Huddart, S. and M. Lang 1996. Employee stock option exercises. An empirical analysis. Journal of Accounting and Economics 21(1):5–43.
- [21] Hull, J. C. and A. White 2004. How to value employee stock options. *Financial Analysts Journal* 60(1):114–19.
- [22] Ingersoll, J. E. 2006. The subjective and objective evaluation of incentive stock options. *Journal of Business* 79(2):453–487.
- [23] International Accounting Standard Board 2004. International financial reporting standard n° 2: Share-based payment: 2004. IASB (London).
- [24] Jennergren, L. P. and B. Näslund 1993. A comment on "valuation of executive stock options and the FASB proposal". Accounting Review 68(1):179–183.
- [25] Johnson, S. and Y. Tian 2000. Indexed executive stock options. Journal of Financial Economics 57(1):35–64.
- [26] Jorgensen, P. L. 2002. American-style indexed executive stock options. *European Finance Review* 6:321–358.
- [27] Kahl, M., J. Liu, and F. A. Longstaff 2003. Paper milionaires: How valuable is stock to a stockholder who is restricted from selling it? *Journal of Financial Economics* 67:385–410.
- [28] Kulatilaka, N. and A. J. Marcus 1994. Valuing employee stock options. *Financial Analysts Journal* 50(6):46–56.
- [29] Lambert, R. A., D. F. Larcker, and R. E. Verrecchia 1991. Portfolio considerations in valuing executive compensation. *Journal of Accounting Research* 29(1):129–149.
- [30] Leung, T. and R. Sircar 2009. Accounting for risk aversion, vesting, job termination risk and multiple exercises in valuation of employee stock options. *Mathematical Finance* 19(1):99–188.
- [31] Sircar, R. and W. Xiong 2007. A general framework for evaluating executive stock options. Journal of Economic Dynamics and Control 31(7):2317–2349.
- [32] Tian, Y. S. 2004. Too much of a good incentive? the case of executive stock options. Journal of Banking and Finance 28:1225–1245.
- [33] Yermack, D. 1998. Companies' modest claims about the value of CEO stock options awards. *Review of Quantitative Finance and Accounting* 10:207–226.

PUBLISHED ISSUES*

WP-AD 2008-01	"Service provision on a network with endogenous consumption capacity" N. Georgantzis, C. Gutiérrez-Hita. March 2008
WP-AD 2008-02	"Capacity restriction by retailers" R. Faulí-Oller. March 2008
WP-AD 2008-03	"The role of the interchange fee on the effect of forbidding price discrimination of ATM Services" R. Faulí-Oller. March 2008
WP-AD 2008-04	"Is bundling anticompetitive?" I. Chioveanu. March 2008
WP-AD 2008-05	"Public transfers to the poor: is Europe really more generous than the US?" M.D.Collado, I. Iturbe, March 2008
WP-AD 2008-06	"Life-cycle portofolio choice: the role of heterogeneity and under-diversification" A.Campanale. March 2008
WP-AD 2008-07	"A new approach for bounding awards in bankruptcy problems" M. Jiménez, M.C. Marco. March 2008
WP-AD 2008-08	"Anti-poverty transfers without riots in Tunisia" C. Muller. May 2008
WP-AD 2008-09	"A multiobjective approach using consistent rate curves to the calibration of a Gaussian Heath–Jarrow–Morton model" A. Falcó, Ll. Navarro, J. Nave. May 2008
WP-AD 2008-10	"Giving it now or later: altruism and discounting" J. Kovarik. July 2008
WP-AD 2008-11	"Specification tests for the distribution of errors in nonparametric regression: a Martingale approach" J. Mora, A. Pérez. July 2008
WP-AD 2008-12	"Optimal two-part tariff licensing contracts with differentiated goods and endogenous R&D" R. Fauli-Oller, J. Sandonis. September 2008
WP-AD 2008-13	"Estimating and forecasting GARCH volatility in the presence of outliers" M.A. Carnero, D. Peña, E. Ruiz. October 2008.
WP-AD 2008-14	"Asset pricing in a general equilibrium production economy with Chew-Dekel risk preferences" C. Campanale, R. Castro, G.L. Clementi. October 2008.

^{*} Please contact Ivie's Publications Department to obtain a list of publications previous to 2008.

WP-AD 2008-15	"Framing effects in public goods: prospect theory ad experimental evidence" I. Iturbe-Ormaetxe, G. Ponti, J. Tomás, L. Ubeda. December 2008.
WP-AD 2008-16	"A parametric control function approach to estimating the returns to schooling in the absence of exclusion restrictions: an application to the NLSY" L. Farré, R. Klein, F. Vella. December 2008.
WP-AD 2008-17	"Second best efficiency in auctions" A. Hernando-Veciana, F. Michelucci. December 2008.
WP-AD 2008-18	"The reliability of self-reported home values in a developing country context" M. González, C. Quitana. December 2008.
WP-AD 2008-19	"Commuting costs and labor force retirement" J. González. December 2008.
WP-AD 2009-01	"Does sex education influence sexual and reproductive behaviour of women? Evidence from Mexico" P. Ortiz. February 2009.
WP-AD 2009-02	"Expectations and frward risk premium in the Spanish power market" M.D. Furió, V. Meneu. February 2009.
WP-AD 2009-03	"Solving the incomplete markets model with aggregate uncertainty using the Krusell- Smith algorithm" L. Maliar, S. Maliar, F. Valli. February 2009.
WP-AD 2009-04	"Employee types and endogenous organizational design: an experiment" A. Cunyat, R. Sloof. February 2009.
WP-AD 2009-05	"Quality of life lost due to road crashes" P. Cubí. February 2009.
WP-AD 2009-06	"The role of search frictions for output and inflation dynamics: a Bayesian assessment" M. Menner. March 2009.
WP-AD 2009-07	"Factors affecting the schooling performance of secondary school pupils – the cost of high unemployment and imperfect financial markets" L. Farré, C. Trentini. March 2009.
WP-AD 2009-08	"Sexual orientation and household decision making. Same-sex couples' balance of power and labor supply choices" S. Oreffice. March 2009.
WP-AD 2009-09	"Advertising and business cycle fluctuations" B. Molinari, F. Turino. March 2009.
WP-AD 2009-10	"Education and selective vouchers" A. Piolatto. March 2009.
WP-AD 2009-11	"Does increasing parents' schooling raise the schooling of the next generation? Evidence based on conditional second moments" L. Farré, R. Klein, F. Vella. March 2009.
WP-AD 2009-12	"Equality of opportunity and optimal effort decision under uncertainty" A. Calo-Blanco. April 2009.

WP-AD 2009-13	"Policy announcements and welfare" V. Lepetyuk, C.A. Stoltenberg. May 2009.
WP-AD 2009-14	"Plurality versus proportional electoral rule: study of voters' representativeness" A. Piolatto. May 2009.
WP-AD 2009-15	"Matching and network effects" M. Fafchamps, S. Goyal, M.J. van der Leij. May 2009.
WP-AD 2009-16	"Generalizing the S-Gini family –some properties-" F.J. Goerlich, M.C. Lasso de la Vega, A.M. Urrutia. May 2009.
WP-AD 2009-17	"Non-price competition, real rigidities and inflation dynamics" F. Turino. June 2009.
WP-AD 2009-18	"Should we transfer resources from college to basic education?" M. Hidalgo-Hidalgo, I. Iturbe-Ormaetxe. July 2009.
WP-AD 2009-19	"Immigration, family responsibilities and the labor supply of skilled native women" L. Farré, L. González, F. Ortega. July 2009.
WP-AD 2009-20	"Collusion, competition and piracy" F. Martínez-Sánchez. July 2009.
WP-AD 2009-21	"Information and discrimination in the rental housing market: evidence from a field experiment" M. Bosch, M.A. Carnero, L. Farré. July 2009.
WP-AD 2009-22	"Pricing executive stock options under employment shocks" J. Carmona, A. León, A. Vaello-Sebastiá. September 2009.
WP-AD 2009-23	"Who moves up the career ladder? A model of gender differences in job promotions" L. Escriche, E. Pons. September 2009.
WP-AD 2009-24	"Strategic truth and deception" P. Jindapon, C. Oyarzun. September 2009.



l vie

Guardia Civil, 22 - Esc. 2, 1° 46020 Valencia - Spain Phone: +34 963 190 050 Fax: +34 963 190 055

Department of Economics University of Alicante

Campus San Vicente del Raspeig 03071 Alicante - Spain Phone: +34 965 903 563 Fax: +34 965 903 898

Website: http://www.ivie.es E-mail: publicaciones@ivie.es